

RESPONSE LETTER

- **Response to reviewer n ° 00506472:** 1) In haemophilia, coronary heart disease occurs at a similar frequency as in the general population (Roger VL, Go AS, Lloyd-Jones DM et al. Heart disease and stroke statistics–2012 update: a report from the American Heart Association. Circulation 2012). I have added the reference and a sentence. 2) We decided to not infuse recombinant FVIII in our patient, because the bleeding risk was very low and he had a high risk of coronary thrombosis. There aren't specific guidelines for the management of angina unstable in these subset of patient, but only consensus documents and few case reports. Many case describes acute coronary syndrome during administration of recombinant FVIII. We have valuated the risks and benefits, in our patient, and we decided to not pretreat, for the very low bleeding risk. We think that the decision should be patient-tailored.
- **Response to reviewer n °00646241:** thank you so much for your comments. I have made some corrections.
- **Response to reviewer n °01277682:** thank you so much!!
- **Response to reviewer n °00225156:** thank you for your comments. I have added the reference of Fefer.